# UCSF UC San Francisco Previously Published Works

## Title

Integrated model of de novo and inherited genetic variants yields greater power to identify risk genes.

## Permalink

https://escholarship.org/uc/item/38m1f9xr

Journal PLoS Genetics, 9(8)

### **Authors**

He, Xin Sanders, Stephan Liu, Li <u>et al.</u>

**Publication Date** 

2013

## DOI

10.1371/journal.pgen.1003671

## **Copyright Information**

This work is made available under the terms of a Creative Commons Attribution License, available at <u>https://creativecommons.org/licenses/by/4.0/</u>

Peer reviewed

# Integrated Model of *De Novo* and Inherited Genetic Variants Yields Greater Power to Identify Risk Genes

Xin He<sup>1</sup>, Stephan J. Sanders<sup>2</sup>, Li Liu<sup>3</sup>, Silvia De Rubeis<sup>4,5</sup>, Elaine T. Lim<sup>6,7</sup>, James S. Sutcliffe<sup>8</sup>, Gerard D. Schellenberg<sup>9</sup>, Richard A. Gibbs<sup>10</sup>, Mark J. Daly<sup>6,7</sup>, Joseph D. Buxbaum<sup>4,5,11,12</sup>, Matthew W. State<sup>2</sup>, Bernie Devlin<sup>13</sup>, Kathryn Roeder<sup>1,3</sup>\*

1 Lane Center of Computational Biology, Carnegie Mellon University, Pittsburgh, Pennsylvania, United States of America, 2 Departments of Psychiatry and Genetics, Yale University School of Medicine, New Haven, Connecticut, United States of America, 3 Department of Statistics, Carnegie Mellon University, Pittsburgh, Pennsylvania, United States of America, 4 Seaver Autism Center for Research and Treatment, Icahn Mount Sinai School of Medicine, New York, New York, United States of America, 5 Department of Psychiatry, Icahn Mount Sinai School of Medicine, New York, New York, United States of America, 5 Department of Psychiatry, Icahn Mount Sinai School of Medicine, New York, New York, United States of America, 6 Analytic and Translational Genetics Unit, Department of Medicine, Massachusetts General Hospital and Harvard Medical School, Boston, Massachusetts, United States of America, 7 Program in Medical and Population Genetics, Broad Institute of Harvard and MIT, Cambridge, Massachusetts, United States of America, 8 Vanderbilt Brain Institute, Departments of Molecular Physiology & Biophysics and Psychiatry, Vanderbilt University, Nashville, Tennessee, United States of America, 9 Pathology and Laboratory Medicine, Perelman School of Medicine, University of Pennsylvania, United States of America, 10 Human Genome Sequencing Center, Baylor College of Medicine, Houston, Texas, United States of America, 11 Department of Genetics and Genomic Sciences, Icahn Mount Sinai School of Medicine, New York, University of Pittsburgh School of Medicine, Pittsburgh, Pennsylvania, United States of America, 13 Department of Psychiatry, University of Pittsburgh School of Medicine, Pittsburgh, Pennsylvania, United States of America, 13 Department of Psychiatry, University of Pittsburgh School of Medicine, Pittsburgh, Pennsylvania, United States of America

#### Abstract

De novo mutations affect risk for many diseases and disorders, especially those with early-onset. An example is autism spectrum disorders (ASD). Four recent whole-exome sequencing (WES) studies of ASD families revealed a handful of novel risk genes, based on independent de novo loss-of-function (LoF) mutations falling in the same gene, and found that de novo LoF mutations occurred at a twofold higher rate than expected by chance. However successful these studies were, they used only a small fraction of the data, excluding other types of de novo mutations and inherited rare variants. Moreover, such analyses cannot readily incorporate data from case-control studies. An important research challenge in gene discovery, therefore, is to develop statistical methods that accommodate a broader class of rare variation. We develop methods that can incorporate WES data regarding de novo mutations, inherited variants present, and variants identified within cases and controls. TADA, for Transmission And De novo Association, integrates these data by a gene-based likelihood model involving parameters for allele frequencies and gene-specific penetrances. Inference is based on a Hierarchical Bayes strategy that borrows information across all genes to infer parameters that would be difficult to estimate for individual genes. In addition to theoretical development we validated TADA using realistic simulations mimicking rare, large-effect mutations affecting risk for ASD and show it has dramatically better power than other common methods of analysis. Thus TADA's integration of various kinds of WES data can be a highly effective means of identifying novel risk genes. Indeed, application of TADA to WES data from subjects with ASD and their families, as well as from a study of ASD subjects and controls, revealed several novel and promising ASD candidate genes with strong statistical support.

Citation: He X, Sanders SJ, Liu L, De Rubeis S, Lim ET, et al. (2013) Integrated Model of *De Novo* and Inherited Genetic Variants Yields Greater Power to Identify Risk Genes. PLoS Genet 9(8): e1003671. doi:10.1371/journal.pgen.1003671

Editor: Scott M. Williams, Dartmouth College, United States of America

Received January 22, 2013; Accepted June 10, 2013; Published August 15, 2013

**Copyright:** © 2013 He et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This work was directly supported by NIH grants R01MH089208 (MJD), R01 MH089025 (JDB), R01 MH089004 (GDS), R01MH089175 (RAG), R01 MH089482 (JSS), R01 MH057881 (BD), R01 MH061009 (JSS), 1U01MH100233-01 (JDB), and 1U01MH100209 (BD). This work was also supported by a grant from the Simons Foundation. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

\* E-mail: roeder@stat.cmu.edu

#### Introduction

The genetic architecture of autism spectrum disorders (ASD) is complex and thought to involve the action of at least hundreds of genes. Yet, despite this complexity, four recent studies [1–4] identified five novel genes affecting the risk for ASD from wholeexome sequencing (WES) of 932 ASD probands. The studies made these discoveries by also sequencing the parents of the probands and thereby discovering a multiplicity of independent Loss-of-Function (LoF) mutations in each of these five genes. The multiplicity is key: due to the rarity of *de novo* LoF events, two or more independent recurrent events in a sample of this size generate more evidence for association than would two LoF variants found in a comparable case and control sample. Thus, even though *de novo* events are rare, these observations provide an excellent signal-to-noise ratio, have proven valuable in the pursuit of reliable signals for genes affecting the ASD risk, and are likely to form the foundation for many studies targeting gene discovery in the future [5].

Note, however, that the multiplicity test is using only a small fraction of all the information collected by a WES study. Many other *de novo* events occur, beyond LoF, and these are ignored. Moreover it completely ignores inherited rare variants within families. And, of course, delineation of rare variants into inherited

PLOS GENETICS

#### **Author Summary**

The genetic underpinnings of autism spectrum disorder (ASD) have proven difficult to determine, despite a wealth of evidence for genetic causes and ongoing effort to identify genes. Recently investigators sequenced the coding regions of the genomes from ASD children along with their unaffected parents (ASD trios) and identified numerous new candidate genes by pinpointing spontaneously occurring (de novo) mutations in the affected offspring. A gene with a severe (de novo) mutation observed in more than one individual is immediately implicated in ASD; however, the majority of severe mutations are observed only once per gene. These genes create a short list of candidates, and our results suggest about 50% are true risk genes. To strengthen our inferences, we develop a novel statistical method (TADA) that utilizes inherited variation transmitted to affected offspring in conjunction with (de novo) mutations to identify risk genes. Through simulations we show that TADA dramatically increases power. We apply this approach to nearly 1000 ASD trios and 2000 subjects from a case-control study and identify several promising genes. Through simulations and application we show that TADA's integration of sequencing data can be a highly effective means of identifying risk genes.

and *de novo* is challenging or impossible for case-control studies. We conjecture that the distribution of variation, whether inherited, *de novo* and from case-control, can be leveraged, in combination with the *de novo* mutations, to maximize the statistical power to detect risk genes.

We propose an integrated model of *de novo* mutations and transmitted variation to address these challenges. We demonstrate that both the number of de novo mutations and the numbers of different types of transmitted variations in family trios (father, mother and an affected child), follow simple distributions dependent on a set of common parameters: mutation rates, relative risks of mutations and population frequency of the variants. This model readily incorporates additional data from case-control studies. The statistical framework of our model enables us to rigorously analyze the genetic architecture of a complex disease, conduct power and sample size analysis, and identify risk genes with higher sensitivity. Through simulations we show that the power of our novel statistical test, called TADA for "transmission and de novo association", is substantially higher than competing tests. Our simulations also provide guidance in planning future studies targeting discovery of genes involved in the risks of complex diseases, henceforth, risk genes.

We demonstrate the benefits of TADA through an extensive study of ASD using published WES data from 932 ASD trios as well as nearly 1000 ASD subjects and matched control subjects from the ARRA Autism Sequencing Consortium (AASC) study [6,7]. Using the model underlying TADA, we estimate there are approximately 1000 genes that play a role in risk for ASD, with an average relative risk of approximately 20 due to LoF in one of these genes. Finally, we identify several potential novel ASD risk genes (genes whose mutations affect the risk of ASD) using TADA and the ASD data.

#### Results

#### Multiplicity test of de novo mutations

For concreteness we start by reviewing the *multiplicity test* to detect risk genes by evaluating the independent recurrence of *de* 

novo mutations in the same gene. The multiplicity test classifies a gene as affecting risk if it sustains d or more recurrent de novo LoF mutations in a sample of N families. Based on computations of expected rates of de novo events as a function of a gene's exonic length and base pair composition [2], a recent study [1] found that  $d \ge 2$  LoF events for N < 1000 is significant evidence to declare a gene as a risk gene (p < 0.05, genomewide). Applying this threshold to data from four ASD family studies [1–4] led to the discovery of five novel genes affecting ASD risk.

A weakness of the multiplicity test is that it produces a single threshold for the entire genome, regardless of the heterogeneity amongst genes in their sizes and base pair composition, and its threshold is a function of sample size, so that the threshold for N = 1,000 is inadequate when the sample increases to N = 10,000. To illustrate the power of the Multiplicity Test and its properties, we performed some simulations using genetic parameters that are described and estimated in the next section.

As demonstrated previously [1], the power for detecting a gene increases monotonically with increasing sample size N and it depends strongly on the gene's mutation rate (Figure 1A). Although the per gene power is relatively low, for a disorder like ASD, more than 60 genes are expected to contain at least two LoF mutations with N = 5000 families (Figure 1B). The corresponding false discovery rate (FDR) is less than 5% for N < 2000 and well below 10% for N as large as 5,000; switching to a threshold of d = 3 to diminish false discoveries leads to a significant loss in power (Figure 1B).

The original treatment of the multiplicity test as requiring a single threshold is simple to adjust. Instead one can compute the p-value for each gene using a Poisson model for the probability of observing  $X_d$  or more recurrent *de novo* events based on the gene's mutation rate. We will call such a test the *De Novo Test*. This test automatically incorporates the number of families and a gene specific mutation rate to determine the likelihood of recurrent *de novo* events.

# Model of *de novo* and inherited mutations in a family design

TADA model is formulated for sequence data from individual genes. Data for the model can come from sequences of trios (unaffected parents and an affected child) and from cases and controls. Given the information from a gene, namely the pattern of de novo mutations and inherited, damaging variants in the affected progeny, the goal is to relate the data with the underlying genetic parameters such as the relative risk of the mutations. In the model, we restrict the class of variation to rare and deleterious mutations acting dominantly and assume subjects can be classified as carrying one of two "alleles", those with a deleterious mutation of this type (a) and those without (A). We put alleles in quotes because, for example, we treat all LoF events in the same gene as a single LoF "allele". Because severe mutations are generally present at very low frequencies in the population (typically <.001), there are effectively two possible genotypes per gene, AA and Aa. If we let q/2 denote the allele frequency of a, then the frequencies of the genotypes AA and Aa in the population are approximately 1-qand q, respectively.

For a trio consisting of unaffected parents and an affected child, there are four likely genotype combinations (Figure 2), of which only three are informative: if both parents are homozygous, a heterozygous child results from a *de novo* mutation; and if one parent is heterozygous, the *a* allele is either transmitted or not. Based on the *de novo* and transmitted alleles, we formulate a likelihood model for the observed data. Let  $\mu$  denote the rate of mutation for the gene being analyzed per generation and



**Figure 1. Properties of the Multiplicity Test.** (A) The probability a risk gene has two or more *de novo* LoF mutations in *N* families (i.e., the power) depends on the mutation rate  $\mu$ . Power per gene of the Multiplicity Test as a function of *N* is shown for 4 mutation rates, which were chosen based on percentiles (25'th, 50'th, 75'th, 90'th) of the distribution of  $\mu$  obtained from the full gene set. (B) The expected number of risk genes discovered by the Multiplicity Test at d = 2 (red, solid) or 3 (blue, dashed) as a function of the sample size *N*. The barplot shows the FDR at d = 2. The simulation assumes 1000 diseases genes out of 18,000, each with relative risk  $\gamma = 20.2$ ; these parameters were estimated in the section on Genetic Architecture of ASD.

doi:10.1371/journal.pgen.1003671.g001

chromosome; let  $\gamma$  denote the genotype relative risk for the genotype Aa; and let f and  $\gamma f$  denote the penetrance of AA and Aa, respectively. Let  $X_d$ ,  $X_t$  and  $X_{nt}$  be the counts of each of the three outcomes (*de novo*, transmitted and nontransmitted, respectively), from a sample consisting of  $N_d$  families. These counts approximately follow Poisson distributions (see Text S1 for derivation):  $X_d \sim \text{Pois}(2\mu\gamma N_d)$ ,  $X_t \sim \text{Pois}(q\gamma N_d)$ , and  $X_{nt} \sim \text{Pois}(qN_d)$ .

For case-control data, counts of genotype Aa in  $N_{\text{case}}$  cases and  $N_{\text{control}}$  controls follow a Poisson distribution with approximate rate parameters  $q\gamma N_{\text{case}}$  and  $qN_{\text{control}}$ , respectively (see Text S1).

From this structure it is apparent that the transmitted counts can be viewed as a type of case-control data with sample size  $N_d$ . Combining data, let  $X_0$  be the total number of Aa in the controls plus the number of transmitted A variants, and let  $X_1$  be the total number of Aa in the cases plus the number of transmitted avariants. It follows that

$$X_d \sim \operatorname{Pois}(2\mu\gamma N_d)$$
  $X_0 \sim \operatorname{Pois}(qN_0)$   $X_1 \sim \operatorname{Pois}(q\gamma N_1)$  (1)

for which  $N_0 = N_{\text{control}} + N_d$  and  $N_1 = N_{\text{case}} + N_d$ . The resulting probability model has three parameters  $(\mu, q, \gamma)$  per gene. For each



**Figure 2.** A probabilistic model for a family trio with an affected child. Genotype probabilities are computed as the marginal probability of parental genotypes times the conditional probability of the child, given the parents. The parameters  $\mu$  and q represent the mutation rate, and the population frequency of the Aa genotype, respectively. Phenotype probabilities for the child, given genotype, are a function of  $f_0$ , the penetrance of the AA genotype, and  $\gamma$  the relative risk of the mutation a. Rate is the (approximate) rate of observing counts  $X_d$ ,  $X_{nt}$  and  $X_t$  from the latter 3 types of trios, respectively.

doi:10.1371/journal.pgen.1003671.g002

gene, the mutation rate per gene  $(\mu)$  can be estimated from its exonic length and nucleotide content [1] and hence this quantity can be treated as known. The statistical problem for each gene is to estimate q and then test if  $\gamma = 1$ .

#### Transmission And De novo Association test: TADA

We conjecture that a more powerful strategy to discover risk genes from family data is to combine the information on de novo and inherited mutations into an unified statistical framework, such as the one we just proposed, which forms the basis for TADA. TADA tests the hypothesis  $H_0: \gamma = 1$  against the alternative  $H_1: \gamma \neq 1$ . A traditional likelihood ratio test will not work well in this setting because one or more of the counts will be zero for many genes, leading to poor maximum likelihood estimates for q and  $\gamma$ . To circumvent this problem we cast TADA in a Hierarchical Bayes (HB) framework, thereby improving estimates of q and  $\gamma$  by pooling information across all genes, but still modeling rates as gene-specific. The underlying assumption is that LoF and severe missense mutations are rare in all genes and hence we can learn about the frequency distribution in a given gene by looking at the distribution across all genes. Likewise, we can learn about how mutations in one gene affect risk by examining the range and distribution of risks across all disease-related genes.

The HB model assumes a fraction  $\pi$  of the genes are associated with the disorder (model  $H_1$ ); the remaining fraction follow the null model (model  $H_0$ ). Under  $H_0$ , the relative risk is constrained ( $\gamma = 1$ ), but under  $H_1$ ,  $\gamma$  is assumed to follow a distribution across risk genes. For both models, the frequency of severe mutations per gene, q, is assumed to vary by gene, with some commonality across the genome. The distributions of  $\gamma$  and q under both models are specified by prior parameters, and we estimate the values of these parameters by maximizing the marginal likelihood of the data (this is known as the Empirical Bayes method, see Methods). Once the prior parameters are estimated, we compute the evidence for  $H_1$ and  $H_0$  for each gene. Specifically, for the *i*-th gene, let  $x_i$  be its data, the evidence for  $H_1$  is defined as:

$$P(x_i|H_1) = \int p(x_i|q_i,\gamma_i)p(q_i|H_1)p(\gamma_i|H_1)dq_id\gamma_i$$
(2)

where  $p(x_i|q_i,\gamma_i)$  is given by Equation 1,  $p(q_i|H_1)$  and  $p(\gamma_i|H_1)$ represent the prior distributions. Unlike the likelihood-based test, the evidence for  $H_1$  is not based on point estimates of q and  $\gamma$ ; instead it integrates out the two parameters. The model evidence of  $H_0$  can be defined similarly, except that  $\gamma$  is fixed at 1. The Bayes factor of any gene is the ratio of  $P(x_i|H_1)$  to  $P(x_i|H_0)$ . The statistical significance of the Bayes factor is given by its p-value, determined empirically by simulating data under the model assuming  $\gamma = 1$  (see Text S1).

Some insights into the relationship to a likelihood-ratio test (LRT) can be gained by examining an approximation of B, the Bayes factor:

$$B = \frac{p(x_i|H_1)}{p(x_i|H_0)} \approx \frac{p(x_i|\hat{q}_i(H_1), \hat{\gamma}_i(H_1))}{p(x_i|\hat{q}_i(H_0), \gamma = 1)}$$
(3)

where the parameters are estimated by Bayesian mean posterior estimators. These parameter estimates are a weighted average of the maximum-likelihood estimate for the i-th gene and the mean of the prior distributions. For example,  $\hat{q}_i(H_0)$  is interpolated between the allele frequency derived from all genes and the genespecific estimate (Figure S1). Thus the Bayes factor is similar to the LRT except that we utilize a refined estimator of the allele frequency.

The model just described is designed for a single type of mutation (say LoF), but it can incorporate multiple types. For different types of mutations, such as LoF and damaging missense mutations, the distributions of  $\gamma$  and q are likely to be different, so we model each type of mutation and estimate the prior parameters separately using the HB framework. Then the total Bayes factor of a gene is the product of the Bayes factor from each type of mutations. In practice, we note that the damaging missense mutations predicted by bioinformatic tools likely contain a number of mutation having no effect on the gene function, thus we introduce an additional model to account for this feature, downweighting the evidence from missense mutations (see Methods).

The TADA method we described can also be used for *de novo* data alone. Basically, we ignore inherited and standing variants, but allow multiple types of *de novo* mutations. The details are not repeated here, but are provided in our supporting Website (see Methods). We call this simplified model, TADA-Denovo, and it is particularly useful for genes with multiple *de novo* events in different categories (e.g. some nonsense and some missense mutations).

#### Genetic architecture of ASD

We use the proposed model to estimate the number of ASD risk genes (k), their average relative risk ( $\bar{\gamma}$ ), and the distribution of the population frequency of the mutations. These estimates yield insight into the genetics of ASD and pave the way for realistic simulations to study the power of statistical tests. Our overall strategy is first to use *de novo* mutations to estimate an approximate range of the parameter values, then use the HB method to refine these estimates using both family and the case-control data.

Consider the *de novo* LoF mutations in N = 932 families [1–4]. These data reveal a total of C = 123 de novo LoF mutations across all genes, and M = 5 multiple-hit genes (at least 2 independent de *novo* LoF events per gene). Our goal is to find values of k and  $\bar{\gamma}$  that best predict the observed counts C and M (Text S1). We assume that the relative risk of an ASD risk gene varies across k genes, with the average relative risk of the LoF mutations equal to  $\bar{\gamma}$ . The mathematics of TADA reveal there is an inverse relationship between k and  $\bar{\gamma}$  (Figure 3A, see Equation 27 in Text S1). For an alternative and more intuitive explanation of why these parameters have an inverse relationship, see the arguments in [2]. For any given value of k, we can compute the expected number of multiple-hit genes; matching the expected with the observed value of M, we estimate the number of ASD risk genes is between 550 to 1000 (Figure 3B). In the next step, we use the HB model to estimate the most likely value of k within this range, and the result is k = 1000 ASD risk genes, with the corresponding relative risk  $\gamma = 20.2$  (see Text S1). These estimates are similar to published results using somewhat different methods [1,2].

We examine evidence for the hypothesis that the population frequency of LoF mutations for ASD risk genes  $(q_1)$  is lower than that for non-risk genes  $(q_0)$  because mutations in ASD risk genes are under stronger negative selection than the average gene. These frequencies are of interest because they have a major influence on the power of association test [8]. We estimate q based on the number of LoF variants in the case-control data from the AASC [7] and the transmitted/nontransmitted data from 641 families (the transmission data are only available for a subset of the 932 families). To obtain the empirical distribution of q across all genes we first count the frequency of the LoF mutations in each gene (Figure 3C); we find a substantial number of genes with 0 LoFs.



**Figure 3. The genetic parameters of ASD.** (A) The relationship between the number of ASD risk genes (k) and the average relative risk ( $\bar{\gamma}$ ). *m* stands for the total number of genes in the human genome, and  $\nu$  for the fold enrichment of the *de novo* LoF mutations in probands vs. siblings (about 2 in our data). (B) The expected number of multi-hit genes (M) in N = 932 families, as a function of the number of ASD risk genes (k). The observed M is 5, and we define the plausible range of k as the values corresponding to M = 4 to 6. The model assumes the relative risks of ASD risk genes follow a gamma distribution with the scale parameter  $\beta$ . The variance of the relative risk ( $\gamma$ ) across genes equals  $\bar{\gamma}/\beta^2$  ( $\bar{\gamma}$  is the average of  $\gamma$  of all ASD risk genes), which limits the range of plausible values for the model. The estimated value of the average  $\gamma$  is approximately 20. (C) For each gene, we compute the empirical allele frequency ( $\hat{q}$ ) of LoFs as the number of LoF variants divided by the sample size. The histogram of the LoF frequencies of all genes is shown. Also shown are the estimated distributions of q under the null (red, solid line) and the alternative (blue, dashed line) models, respectively.

doi:10.1371/journal.pgen.1003671.g003

We next estimate the prior distributions of q under the null and alternative models, respectively, using the HB model and find they provide a good fit to the observed data (Figure 3C, Figure S1). From these analyses the mean of q under  $H_1$ , i.e. the average q for ASD risk genes, is about  $3 \times 10^{-5}$ , significantly smaller than that of non-risk genes,  $6 \times 10^{-4}$  (see Text S1 for a description of how the HB model uses a mixture model to permit estimation of parameters specific to ASD risk genes without actually classifying genes as such.) Notably, while the empirical estimate of q for most genes is 0 (thus not useful for inference), the value of q from the HB model is never equal to 0 due to smoothing.

Using the same procedures we also estimated these parameters for missense mutations that are probably damaging according to the PolyPhen prediction [9] (denoted as Mis3 mutations). Estimates reveal lower risk for these mutations, as expected, and lower q for ASD risk genes compared with non-ASD genes (Table S1).

#### Power analysis by simulation

Equipped with estimates of the genetic parameters, we can simulate genetic data under the model and assess the performance of statistical methods. We compare performance of three tests: De Novo, as described in Section 2.1; TADA, described in Section 2.3; and a "Meta test", which combines two tests, one based on de novo events and the other on inherited variants, via meta analysis. For the meta test we compute the *p*-value from data on inherited variants using a Fisher exact test, treating transmitted/untransmitted events as case-control data; and compute a p-value for de novo events using the De Novo test. Then these p-values are combined using Fisher's method. In all the simulations, different parameters are used to generate the data, yet TADA always uses the same set of parameters derived from the real data, as described previously. Thus these results establish the robustness of TADA under different parameter settings and thus, to some extent, how it should behave for real data.

Because TADA is a novel method, data were first simulated under the null hypothesis of no association to obtain the distribution of the TADA test statistic and its associated *p*-values. The results show that the test is well calibrated and type I error is properly controlled (Figure S2).

Next, data were simulated under the alternative model, using different sample sizes and different combinations of the parameters q and  $\gamma$ , within the range of plausible values estimated in the previous section. This comprehensive simulation showed TADA has superior power relative to the other two tests (Figure S3). In Figure 4, we show a selected portion of the simulation results under the most likely scenarios, reflecting the trade-off between relative risks and allele frequencies, i.e. mutations with high risks are likely to exist in lower frequencies in the population. For a gene with typical parameter values (Figure 4B), the power of the TADA test, at N = 5000, was about fivefold larger than that of the other two tests.

To assess the performance of the tests from a genome-wide analysis, we generated realistic simulated counts based on the estimated genetic parameters for ASD, namely average relative risk of 20 and k = 1000 risk genes, among a total of 18,000 genes sequenced. We focus on false discovery rate (FDR), calibrating the empirical FDR to control at 10%, and estimated power as the number of true discoveries. Results confirmed the advantage of TADA (Figure S4A). For example, at N = 5000, TADA identified more than 200 ASD risk genes at FDR below 10%, while the De Novo and Meta tests identify about 50 and 70 genes at this level of FDR, respectively (cf Figure 1). We performed additional simulations with somewhat different procedures to demonstrate the robustness of these findings. In one experiment, we simulated data under the average relative risk of 10, instead of 20, while TADA still uses the relative risk of 20. The power of all methods was significantly reduced, as expected, yet TADA still performed better than both de novo test and the simple meta-analysis (Figure S4B). In another experiment, the simulation procedure incorporated the possible dependency between the LoF frequency of a gene (q) and its relative risk  $(\gamma)$ , based on simple mutation-selection balance: the two were not sampled independently, but rather the frequency was inversely proportional to the risk (see Methods). Despite this change of simulation model, the results were virtually identical to those from earlier simulations (Figure S4C).



**Figure 4. The power per gene of competing tests.** The results of three tests are shown: novo (red), meta (blue), and TADA (purple). Results are shown for various values of *N*,  $\gamma$  and *q* with type I error fixed at 0.001. Parameter values are chosen to cover plausible parameter values according to our model estimation: (A)  $\gamma = 15$ ,  $q = 1 \times 10^{-4}$ ; (B)  $\gamma = 20$ ,  $q = 5 \times 10^{-5}$ ; and (C)  $\gamma = 25$ ,  $q = 2.5 \times 10^{-5}$ . doi:10.1371/journal.pgen.1003671.g004

# Analysis of data to identify genes affecting the risk of ASD

The data we used were all reported *de novo* mutations from 932 ASD families [1–4]; transmitted mutations from 641 of these families; and case-control data from the AASC, consisting of 935 ASD subjects and 870 controls [7]. Each missense mutation was classified into a category of damage to the protein based on its predicted effect on the coding sequence using PolyPhen2 [9]: benign (Mis1); possibly damaging (Mis2); and probably damaging (Mis3). Note that *de novo* LoF mutations occurred at about two-fold enriched rate in the probands relative to the unaffected siblings (Figure 5A, Table S2). The rate for *de novo* Mis3 was also higher in probands than siblings, but the difference was not as striking. There is essentially no difference in probands and siblings for other types of mutations. We thus applied the TADA method to the LoF and Mis3 mutations.

The overall inflation of the results due to population stratification is negligible: a modified [7] genomic control factor [10]  $\lambda_q = 1.02$  (see Text S1). There is significant enrichment of genes with low *p*-values compared with random expectation (Figure 5B): 244 genes have *p* - value < 0.01, 64 more than expected under the null model. There is an intriguing coincidence in the excess of small *p*-values - namely that it is very similar to the excess number of genes with single-hit *de novo* LoF events in ASD subjects



**Figure 5. Application of TADA to the genetic data of ASD.** (A) *De novo* LoF and "probably damaging" missense mutations are enriched in ASD probands (red) compared with unaffected siblings (blue), based on a comparison including all trio and quad families. The other types of missense mutations are not enriched. To make the numbers comparable, the number of mutations in siblings is scaled by a constant multiplier (214/124) so that the numbers of silent mutations is equal in probands and in siblings. The annotations of missense mutations are based on PolyPhen. (B) Q-Q plot (log. scale) of the *p* values for all genes in the ASD dataset based on a combined analysis of LoF and severe missense mutations. doi:10.1371/journal.pgen.1003671.g005

Table 1. Top predicted ASD risk genes from the TADA analysis of combined ASD data (de novo, inherited and case-control).

Gene	Loss of func						
	De novo	Transmitted	Nontransmitted	Case	Control	<i>p</i> <sub>dn</sub>	P <sub>TADA(LoF)</sub>
KATNAL2	2	1	0	4	0	3.1×10 <sup>-6</sup>	2×10 <sup>-7</sup>
CHD8	2	0	0	3	0	$9.5 \times 10^{-5}$	$2.4 \times 10^{-6}$
LMCD1	0	2	0	0	0	1	0.067
S100G	1	0	0	3	0	0.00042	$1.6 \times 10^{-5}$
DYRK1A	2	0	0	0	0	8.6×10 <sup>-6</sup>	$4.3 \times 10^{-6}$
PPM1D	1	0	0	2	0	0.0032	0.00023
SCN2A	2	0	0	0	0	5.9×10 <sup>-5</sup>	$2.8 \times 10^{-5}$
CUL3	1	0	0	3	0	0.004	0.00013
DEAF1	0	2	0	1	0	1	0.031
BANK1	0	1	0	4	0	1	0.0064
POGZ	2	0	0	0	0	3×10 <sup>-5</sup>	$1.4 \times 10^{-5}$
WDR55	0	1	0	0	0	1	0.18
FAM91A1	1	0	0	0	0	0.0046	0.0019
COL25A1	1	0	0	5	0	0.0034	2.3×10 <sup>-5</sup>

Gene	De novo	Transmitted	Nontransmitted	Case	Control	<i>P</i> TADA
KATNAL2	0	2	3	4	5	1.5×10 <sup>-6</sup>
CHD8	0	4	6	9	9	$1.3 \times 10^{-5}$
LMCD1	0	4	0	9	0	$1.7 \times 10^{-5}$
S100G	0	0	0	0	0	2.1×10 <sup>-5</sup>
DYRK1A	0	0	4	4	1	5.6×10 <sup>-5</sup>
PPM1D	0	0	0	2	0	7.9×10 <sup>-5</sup>
SCN2A	1	8	7	5	5	8.4×10 <sup>-5</sup>
CUL3	0	0	0	1	0	8.6×10 <sup>-5</sup>
DEAF1	0	1	0	8	0	0.0001
BANK1	0	7	0	6	2	0.00011
POGZ	0	4	1	3	5	0.00012
WDR55	1	0	0	6	0	0.00012
FAM91A1	0	12	1	2	2	0.00016
COL25A1	0	5	3	4	4	0.00016

The  $p_{dn}$  column shows the *p*-values using the *De Novo Test* from the *de novo* LoF mutations alone. The  $p_{TADA(LoF)}$  column shows the *p*-values from the TADA test using all LoF data. The  $p_{TADA}$  column shows the *p*-values from the TADA test using both LoF and Mis3 data. The star symbols mark the double-hit genes that were reported in earlier publications. C1orf95 also has *q*-value<.2, however this signal is based entirely on 11 identical Mis3 variants in cases and 0 in controls. This allele is common in African populations [40]. While the AASC sample is of European ancestry, a portion of it, largely from Portugal, carries some sub-Saharan alleles [7]. Thus, this signal is likely due to population substructure. Similarly, the 3 LoF variants seen in S100G are copies of a splice variant that is common in African populations.

doi:10.1371/journal.pgen.1003671.t001

compared to their unaffected siblings [1]. Notably the large tail in the QQ plot is largely driven by the *de novo* LoF events, and appears to reflect true signal instead of inflation.

We control for the multiple hypothesis testing using the Benjamini-Hochberg procedure [11]. Fifteen genes meet the criteria of a False Discovery Rate less than 20% (Table 1, see Table S3 for the complete results). The list includes all five genes with two *de novo* LoF mutations, as well as several novel genes that are promising candidates for ASD based on existing evidence. For the novel predictions, the *p*-values from the *de novo* data alone are far from achieving genome-wide significance (the  $p_{dn}$  column in Table 1) and would be impossible to identify without combining the *de novo*, transmitted and case-control data.

The results of TADA generally depend on the estimates of the mutation rates of the genes, as well as the Bayesian prior parameters of the model. We perform additional analyses to study how sensitive the results are to these parameters. Based on our findings, we choose several genes from Table 1 for this investigation. Although the error of mutation rate estimation is likely small [1], we vary the mutation rate of each gene: from 1/2 of the estimated rate to twice the rate. As expected, the *p*-value increases as the mutation rate increases, although overall the impact is modest (Figure S5A). Next we vary the Bayesian prior parameter,  $\bar{\gamma}$ , which represents the average relative risk over all risk genes, from 10 to 20. The *p*-values from TADA are even less sensitive to this parameter (Figure S5B).

#### Discussion

For disorders like ASD, recent results show that detection of de novo LoF events can be a powerful means of discovering novel risk genes [1-4]. Yet de novo events are relatively rare, roughly one per exome, and de novo LoF events even more so, and thus many families must be assessed to identify multiple de novo LoF events in the same gene. To make the most of this experimental design, we develop a new statistical approach, TADA, that utilizes both transmitted and de novo variants from nuclear families and casecontrol data to determine genetic association. TADA builds on the simple multiplicity test, which relies on recurrent de novo events, but it creates a full analytical framework to incorporate all of the information on the distribution of rare variation. The result is a test with greater power. Our test achieves its good performance properties by providing an analytic framework that links the observed pattern of *de novo* mutations with the underlying genetic parameters, such as the relative risk conveyed by such mutations. In addition to analyzing data for novel gene discovery, this framework can be used to analyze the power of a test and predict the required sample size to attain sufficient power for future investigations. Moreover, by using empirical Bayes methods, TADA refines estimates of allele frequencies of the damaging mutations by using the full genome to estimate these quantities. This approach increases the information in the transmitted variants in each gene considerably and yet maintains good control of false discoveries.

Association studies evaluating cases and controls have been a common design for identifying variation affecting risk for complex diseases. It has proven successful for identifying common variation affecting risk, after sufficient samples had been amassed to ensure variation having modest impact on risk could be detected [12]. Common variants surely play a role in ASD [13,14], but the effect sizes are small [15] and it will be challenging to detect individually-significant SNPs. Indeed virtually every discovery for ASD risk genes traces to rare and *de novo* variants [1–4,16–20].

As the cost of sequencing drops, genetic research increasingly focused on the role of rare variants in complex diseases such as ASD, but the sample size has been limited and so has the yield of such studies. For a sample of nearly 1000 ASD case and well matched controls the ARRA ASD sequencing consortium (AASC) found no significant associations [7], except for variation acting recessively [6]. These results comport with studies of other disorders and suggest that large sample sizes will be required to achieve good power in rare variant association studies [21]. Arguably a fundamental difficulty is that most of the mutations with large effects tend to be under strong negative selection, existing at very low frequencies in the population [22]. Variants that occur with greater frequency often have smaller effect on the phenotype, reducing the power of gene-based test statistics.

Our analysis provides insight into some advantages of *de novo* over case-control studies, especially for LoF events. The *de novo* test gains power because the mutation rate for genes can be estimated accurately from supplementary sources, and need not be estimated as part of the statistical procedure. Because of the low mutation rate, the number of *de novo* LoF events expected by chance is very small, and thus we could attach high statistical significance to any gene with more than one independent LoF mutation. While a single *de novo* LoF event is certainly not definitive evidence, it can put a gene on the short list as a risk gene – for ASD, it is more likely than not an ASD risk gene. In contrast, for case-control data, we require an estimate of the allele frequency q under the null hypothesis. When the mutant allele is very rare (as for ASD risk

genes), a very large sample is required to ensure that this frequency is indeed small.

Another feature of observed *de novo* mutations is that they have not been subject to the force of purifying selection, which plays a key role in shaping the pattern of standing variation. Therefore it is likely that *de novo* mutations, especially LoF mutations, have stochastically larger effect sizes than rare variation transmitted for generations, because selection tends to drive down allele frequencies of variants having large effects on reproductive success. Moreover, allele frequency is inversely tied to power, critical for any experimental design. Therefore studies utilizing *de novo* variation can have distinct advantages, in terms of power, relative to those that do not.

By simulations we demonstrate that the power of TADA is higher than tests based solely on de novo events or standard metaanalysis that combines p-values from de novo and inherited data (transmission or case/control). There are two explanations for this gain of power. First, TADA's hierarchical model uses the information in the case-control (or transmission) data more efficiently than the standard hypergeometric or trend test. One important property of LoF mutations, compared to less severe functional variants, is their rarity in the population (Figure 3C). TADA, which is similar in spirit to a Poisson test of rare events, is able to exploit the rarity of these damaging events by estimating the distribution of LoF alleles across the exome (see Figure S1B), whereas the other methods cannot. Second, because damaging de novo mutations are rare, most genes will not harbor them even when thousands of cases have been sequenced. For such genes, using Fisher's method to combine the de novo p-value, which will be close to 1, with the *p*-value from the case-control data penalizes the overall test statistic. In contrast, the Bayesian approach uses de novo events when they are informative and disregards the de novo data when they are uninformative; the Bayes factor from de novo in such cases would be close to 1, making little contribution to the gene's total Bayes factor.

We estimate that there are about 1,000 ASD risk genes with average relative risk about 20. In a recent paper using the same *de novo* data, the number of ASD risk genes (k) was estimated at 370 [4]. In that paper, the expected number of genes with recurrent LoF events was derived as a function of k, and equating it to 5 (the observed number), produced the solution that k=370. The analysis made the implicit assumption that all ASD risk genes are equally likely to sustain multiple *de novo* LoF events. In Text S1 we show, using Jensen's Inequality, that the non-uniform distribution of the mutation rates and the relative risks among the ASD risk genes leads to a significant under-estimation of k, explaining the discrepancy between our results and those of Iossifov et al. [4].

When applied to ASD data, TADA predicts a number of novel ASD risk genes (Table 1), as well as supporting results for known ASD risk genes. For some of the newly implicated genes it is straightforward to garner other supporting evidence for their role in ASD. *S100G* is a downstream target of *CHD8*, a key transcriptional regulator often disrupted in ASD subjects [23]. *CUL3* plays a critical role in neurodevelopment [24,25] and in particular regulates synaptic functions [26]. A recent study identified an additional *de novo* protein-changing mutation in *CUL3* in ASD probands [27], replicating our finding here. *COL25A1*, a brain-specific collagen, was implicated in risks for Alzheimer's disease [28] and antisocial personality disorder [29].

Inspection of other genes slightly below our chosen FDR threshold reveals several more interesting genes that likely play some role in ASD (all ranked among the top 25, see Table S3). *TBR1*, a transcription factor critical in brain development, regulates several known ASD risk genes [30]. A recent study has

identified recurrent *de novo* disruptive mutations in *TBR1* in ASD subjects [23]. *MED13L*, a component of the Mediator Complex, is intriguing because of its role in Rb/E2F control of cell growth [31] and the fact that RB/E2F plays a key role in neurogenesis [32] and neuronal migration [33]. Recently *MED13L* has been associated with risk for schizophrenia [34]. *NFIA* is a member of the NFI transcription factor family, thought to have a neuroprotective role [35], and *NFIA*-knockout mice display profound defects in brain development [36].

Genotyping/sequencing errors can introduce biases in data analyses, especially those for family data [37,38] and for combining data across multiple heterogeneous studies [39]. Our analyses are likely robust to these possible biases because the variant calls were all carefully evaluated: (i) all de novo mutations described previously [1-4] and analyzed here, a total of 122 LoF and 314 damaging missense mutations, have been validated by previous studies; (ii) the case-control data have been carefully harmonized to minimize batch effects by using stringent quality control filters [7]; and (iii) for the case-control data, all variant calls in two genes (CHD8 and SCN2A) have been evaluated by Sanger sequencing and 20 out of 20 validate, further supporting the quality of the variant calls in the case-control data. When the sensitivity of calling minor variants is low (under-calling), this may create an under-transmission bias in family-based test statistics; however, TADA is effectively a one-sided test of the adverse effect of the minor allele. As such, TADA is only powered to detect risk variants that are over-transmitted and thus bias due to undertransmission is not a significant concern. Nonetheless, data quality is always an important concern, and can change over time in subtle ways [37,38], making high-quality filters and validation of de novo events critical for good data analyses. It is possible that TADA would benefit by modeling measurement errors and this will be a topic for future research, when the error structure in the data is better understood.

While much of our focus has been on ASD data and the genetic architecture of ASD, TADA has utility beyond the genetics of ASD. For example, we would expect TADA to be useful for gene discovery by the analysis of data from any genetic disorder or disease for which *de novo* mutations play a substantive role in risk. Early onset diseases and disorders are obvious candidates for the use of TADA, as are disorders such as schizophrenia and congenital heart disease. Indeed there are a plethora of human diseases for which *de novo* mutations account for at least a small fraction of risk, even diseases that onset in mid-life such as cardiovascular disease. Because TADA is based on a general theoretical framework for combining rare variation found in exons of genes, we predict that its logic can have even broader applications than simply the analysis of single genes for their association with disease.

#### Methods

#### Sequence data

We combined exome sequence data from four recent studies of ASD, covering 932 families [1–4]. Detailed information about study design, including family structure (simplex versus multiplex), ascertainment, and DNA source (blood versus cell line), can be found in the Supplements of these papers. The *de novo* mutations, including both single nucleotide variants (SNVs) and indels, were identified as described in the original papers. The transmitted and non-transmitted variants were extracted from 641 of these families (see Text S1 for details on data processing). We excluded all common variants from the analysis, defined as those present at >1% population frequency in the Exome Sequencing Project

(ESP) controls and/or the 1,282 parents [40]. Only SNVs were called for the transmission data, indels were not identified. We also included case/control data from the ARRA ASD Sequencing Consortium, consisting of 935 ASD subjects of European ancestry and 870 controls of ancestry similar to cases, selected from the NIMH repository (see complete information on study design in the supplement of Liu et al [7]). The SNVs and indels in the case/ control were called as described in [7]. Each mutation/variant in the combined data was classified into different categories, based on its predicted effect on the protein function, according to the program PolyPhen2 [9]. In this study we focused on (1) LoF mutations, defined as nonsense mutations, mutations in splice sites or frameshift indels: and (2) mutations classified as "probably damaging" to protein function by PolyPhen2 (Mis3). We also removed all genes with more than 10 LoF events in the control samples (166 genes in total) from the analysis, as these genes are unlikely to be related to ASD.

#### Mutation rate estimation

For each gene, the total rate of base pair substitutions was estimated using a probability model taking the gene length and base content into account [1]. To estimate the rate of a specific type of mutation (LoF or Mis3) of a gene, we multiplied the gene-level mutation rate and the proportion of that type of mutation. The proportion of LoF or Mis3 mutations was estimated from the data of unaffected siblings in the ASD families (Table S2). In these siblings, there were 461 single-nucleotide variants (SNVs) and 34 LoF variants, thus the LoF fraction was 34/461 = 0.074. Similarly the Mis3 fraction was calculated as 147/461 = 0.32.

#### TADA model and the statistical test

Two hypotheses were compared,  $H_1 : \gamma_i \neq 1$  versus  $H_0 : \gamma_i = 1$ , for each gene. For most genes, the number of LoF mutations either transmitted or not (or in cases and controls) was generally very small and often 0, leading to a naive estimate of  $\hat{q} = 0$  and creating a challenge for a likelihood-based test. To refine inference we took an Empirical Bayes approach and developed a hierarchical Bayes model for the data (Figure 6). We estimated the prior parameters in the model by maximizing the marginal likelihood. The hierarchical model assumed a fraction  $\pi$  of the genes was



**Figure 6. Bayesian hierarchical model of TADA.** A fraction  $\pi$  of the genes are associated with the phenotype under investigation and follow model  $H_1$ , and the remainder follow model  $H_0$ . The prior distribution of gene-specific parameters, relative risk ( $\gamma_i$ ) and allele frequency ( $q_i$ ), can vary under the competing models,  $H_1$  or  $H_0$ . Priors are specified by the hyperparameters,  $\phi_1$  and  $\phi_0$ , respectively, which are estimated from the data. Counts of events for the *i*-th gene follow a Poisson distribution, parameterized by  $\gamma_i$  and  $q_i$  under  $H_1$ , and  $q_i$  under  $H_0$ .

doi:10.1371/journal.pgen.1003671.g006

associated with the disorder (model  $H_1$ ) and the remaining fraction followed the null model ( $H_0$ ). Under  $H_0$ , we assumed  $\gamma = 1$  for all genes and q followed a Gamma( $\bar{q}_0 v_0, v_0$ ) distribution (we parameterized the distribution so that its mean was  $\bar{q}_0$ ). The scaling parameter of the Gamma distribution ( $v_0$ ) played the role of a precision parameter or pseudo count; the bigger  $v_0$  the more similar q was estimated to be across genes. Under  $H_1$ , we assumed  $\gamma_i$  of the *i*-th risk gene follows a Gamma( $\bar{\gamma}\beta,\beta$ ) distribution and  $q_i$ follows a Gamma( $\bar{q}_1 v_1, v_1$ ) distribution.

Let  $\phi_1 = (\bar{\gamma}, \beta, \bar{q}_1, \nu_1)$  be the prior parameters of  $H_1$ , and  $\phi_0 = (\bar{q}_0, \nu_0)$  be those of  $H_0$  (they are also called hyperparameters). The counts for the *i*-th gene,  $x_i = (x_i^1, x_i^0, x_i^d)$ , follow Poisson distributions parameterized by  $\gamma_i$  (1 for non-risk genes) and  $q_i$ , as defined in Equation 1.

The marginal likelihood of the *i*-th gene under either model,  $P(x_i|H_1)$  and  $P(x_i|H_0)$ , is given by:

$$P(x_i|H_0) = \int p(x_i|q_i,\gamma_i=1)p(q_i|H_0)dq_i,$$
 (4)

$$P(x_i|H_1) = \int p(x_i|q_i,\gamma_i)p(q_i|H_1)p(\gamma_i|H_1)dq_id\gamma_i.$$
 (5)

The marginal likelihood of all the data, as a function of the hyperparameters  $(\phi_1, \phi_0)$ , is

$$P(x|\phi,\phi_0) = \prod_{i=1}^{n} [\pi P(x_i|H_1) + (1-\pi)P(x_i|H_0)].$$
(6)

We assume the proportion of risk genes,  $\pi$ , is known (in our analysis of ASD data, this is obtained by the estimated value of k, the number of ASD risk genes, see Section 2.4). The hyperparameters can then be found by maximizing this marginal likelihood function. Once we have the estimated values of  $\phi_1$  and  $\phi_0$ , we compute the Bayes factor of any gene:

$$B_i = \frac{P(x_i|H_1)}{P(x_i|H_0)}.$$
(7)

The *p*-values of the observed Bayes factors are calculated by sampling the null distribution according to Equation 1 (see Text S1).

#### TADA for multiple types of mutations

When analyzing multiple types of mutations (LoF and Mis3 in our analysis of ASD data), we assumed the data for each type of mutation were independent of each other, and hence we estimated the prior parameters for each type of mutation separately. The Bayes factor of a gene is defined as the product of the Bayes factor for each type of mutation. For these ASD data, the Mis3 mutations are likely to be a mixture of those causing protein-damaging changes and those having no real effects on the protein function. We thus computed the joint Bayes factor of the gene using this equation:

$$B = B_{\text{LoF}}[w \cdot B_{\text{Mis3}} + (1 - w)]; \qquad (8)$$

we used w = 0.55 in our ASD analysis (see Text S1).

#### Simulation procedure

Our simulation procedure generated data using the estimated genetic parameters of the LoF mutations of the ASD risk genes (Text S1). For our initial simulations, we compared the power of several statistical tests, at the single gene level, under various combinations of parameter values. We set the mutation rate as the mean mutation rate of the LoF mutations of all human genes,  $1.7 \times 10^{-6}$ . The parameters  $\gamma$  and q were chosen according to their estimated mean values:  $\gamma = 15,20,25$ , and  $q = 5 \times 10^{-5}, 1 \times 10^{-4}, 2 \times 10^{-4}$ . We compared the power of the three tests under type I error 0.001.

For the second set of simulations, we assessed the performance of the three tests in the genomewide setting. Specifically, from among 18,000 genes in the human genome, we first randomly sampled k = 1000 risk genes and the rest were assumed to be unrelated to disease (we used the estimated mutation rates of all genes to make this simulation realistic). For a risk gene and a LoF mutation, the effect size parameter  $\gamma$  was sampled from the distribution  $\gamma \sim \text{Gamma}(20,1)$ . Its population frequency parameter q was sampled from the distribution, Gamma(0.5,10000). For a non-causal gene, its relative risk  $\gamma = 1$ , and the frequency parameter q was sampled from the distribution Gamma(0.5,500). The simulation procedure then generated, for each gene, the number of *de novo* mutations  $(X_d)$ , the number of transmitted variants  $(X_1)$  and the number of nontransmitted variants  $(X_0)$ , according to Equation 1.

We ran the three statistical tests, as described in the text, on the simulated data from all genes. At various significance levels, we calculated the number of true discoveries  $(M_1)$ , i.e. the number of diseases genes whose test statistic reached significance level  $\alpha$ . We chose the value of  $\alpha$  so that FDR is less than 0.1, and reported  $M_1$  at this value of  $\alpha$  (see Text S1 for our procedure for controlling FDR in the simulations.)

In additional simulations, we varied the basic procedure just described. In one setting, the average relative risk was set to 10 instead of 20, i.e.,  $\gamma$  of a risk gene was sampled from the distribution Gamma(10,1). In another setting, instead of sampling q and  $\gamma$  of each risk gene independently, we modeled the two as dependent. Specifically, for the *i*-th risk gene, let  $\gamma_i$  and  $q_i$  be the relative risk and the LoF frequency, respectively. First sample  $\gamma_i$  from Gamma(20,1), then determine  $q_i$  according to a simple mutation-selection balance:  $q_i = \mu_i/\delta$ , in which  $\mu_i$  is the mutation rate and  $\delta$  is a constant. To determine the value of  $\delta$ , we plugged in the mean values of q,  $\mu$  and  $\gamma$  in the above equation and solve  $\delta = 0.0017$ .

#### Software

TADA software is available as an R package at http://wpicr. wpic.pitt.edu/WPICCompGen/. The package also includes TADA-Denovo, the simplified version of TADA, that analyzes only *de novo* data.

#### **Supporting Information**

**Figure S1** Bayesian estimation of the frequency parameter q. (A) The observed LoF counts (red) of all genes, vs. the simulated counts (blue). For simulation of one gene, we first sample q from the estimated prior distribution of q under  $H_0$ , and then generate the count data under this q according to the Poisson model (Equation 1 of the text). The procedure is repeated for all genes, and the resulting barplot is provided along with the distribution of the observed data. Note that we did not use the distribution  $q|H_1$ , as most of the genes are not disease-related. (B) The Bayesian hierarchical model estimation of the allele frequency q of LoF variants. The blue circles show the observed frequencies of 10 different genes, which are also maximum likelihood estimates (MLE). The red circle shows the average q over all genes (prior

mean). The Bayesian posterior mean estimates are the weighted average of the MLE and the prior mean (the intersection of the dashed line and solid lines), with weight w (0.20 in this example). (TIF)

**Figure S2** Typical Q-Q plots under the null distribution of the TADA test statistic. We simulate n = 10000 genes under the null model, with mutation parameter  $\mu = 1.7 \times 10^{-6}$  (the mean LoF mutation rate of all human genes), and q varying from from  $3 \times 10^{-4}$  to  $3 \times 10^{-3}$  (the average q of non-autism genes is about 0.001) and number of family trios (*N*) varying from 1000 to 3000. The TADA model is applied to each of 9 simulated datasets to obtain the *p*-values and resulting Q-Q plots. Although there is normal variation in these samples, most follow the expected null distribution fairly closely.

(TIF)

**Figure S3** The power of the *de novo* test (red), the meta test (blue) and TADA (purple) at type I error 0.001, under various values of N,  $\gamma$  and q.

(TIF)

**Figure S4** The number of discovered disease genes as a function of sample size at FDR equal to 10%. We compare power for a test relying on only *de novo* events (*De novo Test*, red), a test combining *p*values from *de novo* and transmitted data by Fisher's method (Meta test, blue), and the joint likelihood-based analysis (TADA test, purple). Results from three different simulations are shown. (A) Simulation using the estimated ASD parameters (the average relative risk  $\bar{\gamma} = 20$ ). (B) Simulation assuming  $\bar{\gamma} = 10$ . (C) Simulation under the inverse-relationship between the LoF frequency (*q*) and the relative risk ( $\gamma$ ) for each risk gene. (TIF)

**Figure S5** Sensitivity analysis of TADA for four selected genes. (A) For each gene, suppose  $\mu$  is its (estimated) mutation rate, we let TADA use a different rate, ranging from  $\mu/2$  to  $2\mu$ , and the resulting *p*-values are shown. (B) We vary the prior parameter  $\bar{\gamma}$  (the average relative risk of all risk genes) of TADA from 10 to 20, and compute the TADA *p*-values. (TIF)

**Table S1** Parameters from Hierarchical Bayes estimation. The LoF and damaging missense (Mis3) mutations of ASD genes have high relative risks, and appear to be under stronger purifying selection than non-ASD genes.

(PDF)

**Table S2** The statistics of the *de novo* mutations in autism probands and unaffected siblings. The missense labels are based on predictions from PolyPhen2. Missense1–3 correspond to "benign", "possibly damaging" and "probably damaging"

#### References

- Sanders SJ, Murtha MT, Gupta AR, Murdoch JD, Raubeson MJ, et al. (2012) De novo mutations revealed by whole-exome sequencing are strongly associated with autism. Nature 485: 237–241.
- Neale BM, Kou Y, Liu L, Ma'ayan A, Samocha KE, et al. (2012) Patterns and rates of exonic *de novo* mutations in autism spectrum disorders. Nature 485: 242– 245.
- O'Roak BJ, Vives L, Girirajan S, Karakoc E, Krumm N, et al. (2012) Sporadic autism exomes reveal a highly interconnected protein network of *de novo* mutations. Nature 485: 246–250.
- Iossifov I, Ronemus M, Levy D, Wang Z, Hakker I, et al. (2012) De novo gene disruptions in children on the autistic spectrum. Neuron 74: 285–299.
- Veltman JA, Brunner HG (2012) De novo mutations in human genetic disease. Nat Rev Genet 13: 565–575.
- Lim ET, Raychaudhuri S, Sanders SJ, Stevens C, Sabo A, et al. (2013) Rare complete knockouts in humans: population distribution and significant role in autism spectrum disorders. Neuron 77: 235–242.

mutations, respectively. The last row is the counts of frameshift indels.

(PDF)

**Table S3** The complete prediction results of TADA. The "mut.rate" column shows the estimated mutation rate of the genes. For each of the two types of mutations, LoF and mis3 (severely damaging), five counts are shown, including the number of *de novo* mutations, the numbers of transmitted and non-transmitted variants, and the number of variants in cases and controls. The  $p_{dn}$  column shows the *p*-values using the *De Novo Test* from the *de novo* LoF mutations alone. The  $p_{TADA(LoF)}$  column shows the *p*-values from the TADA test using all LoF data. The  $p_{TADA}$  column shows the *p*-values from the TADA test using both LoF and Mis3 data. The last column shows the *q*-value of  $p_{TADA}$  after Benjamini-Hochberg correction of multiple testing. (XLSX)

**Text S1** Supplementary methods explaining the details of TADA, and our analysis of ASD data. (PDF)

#### Acknowledgments

We are grateful to all of the families participating in the Simons Foundation Autism Research Initiative (SFARI) Simplex Collection (SSC) and in the AARA autism sequencing consortium (AASC) study. We thank the SSC principal investigators A. L. Beaudet, R. Bernier, J. Constantino, E. H. Cook Jr, E. Fombonne, D. Geschwind, D. E. Grice, A. Klin, D. H. Ledbetter, C. Lord, C. L. Martin, D. M. Martin, R. Maxim, J. Miles, O. Ousley, B. Peterson, J. Piggot, C. Saulnier, W. Stone, C. A. Walsh and E. Wijsman and the coordinators and staff at the SSC sites for the recruitment and comprehensive assessment of simplex families; the SFARI staff, in particular M. Benedetti, for facilitating access to the SSC; Prometheus Research for phenotypic data management and Prometheus Research and the Rutgers University Cell and DNA repository for accessing biomaterials. We thank Thomas Lehner (NIMH) and Adam Felsenfeld (NHGRI) for their support and contribution to the AASC study. We acknowledge the clinicians and organizations that contributed to samples used in this study, including the Autism Genetics Resource Exchange, the institutions of the Boston Autism Consortium, and The Autism Sequencing Collection (TASC) centers. Recruitment of TASC subjects was supported by Autism Speaks. J.D.B., B.D., M.J.D., R.A.G., G.D.S. and J.S.S. are lead investigators in the Autism Sequencing Consortium (ASC). The ASC is comprised of groups sharing massively parallel sequencing data in autism.

#### **Author Contributions**

Conceived and designed the experiments: XH BD KR. Performed the experiments: XH SJS LL SDR. Analyzed the data: XH SJS LL KR. Contributed reagents/materials/analysis tools: XH SJS LL SDR ETL JSS GDS RAG MJD JDB MWS BD KR. Wrote the paper: XH SJS BD KR.

- Liu L, Sabo A, Neale BM, Nagaswamy U, Stevens C, et al. (2013) Analysis of Rare, Exonic Variation amongst Subjects with Autism Spectrum Disorders and Population Controls. PLoS Genet 9: e1003443.
- Bansal V, Libiger O, Torkamani A, Schork NJ (2010) Statistical analysis strategies for association studies involving rare variants. Nat Rev Genet 11: 773– 785.
- Adzhubei IA, Schmidt S, Peshkin L, Ramensky VE, Gerasimova A, et al. (2010) A method and server for predicting damaging missense mutations. Nat Methods 7: 248–249.
- Devlin B, Roeder K (1999) Genomic control for association studies. Biometrics 55: 997–1004.
- Benjamini Y, Hochberg Y (1995) Controlling the false discovery rate: A practical and powerful approach to multiple testing. Journal of the Royal Statistical Society, Series B 57: 289–300.
- 12. Visscher PM, Brown MA, McCarthy MI, Yang J (2012) Five years of gwas discovery. Am J Hum Genet 90: 7–24.

- Klei L, Sanders SJ, Murtha MT, Hus V, Lowe JK, et al. (2012) Common genetic variants, acting additively, are a major source of risk for autism. Mol Autism 3: 9.
- Anney R, Klei L, Pinto D, Regan R, Conroy J, et al. (2010) A genome-wide scan for common alleles affecting risk for autism. Hum Mol Genet 19: 4072–82.
- Devlin B, Melhem N, Roeder K (2011) Do common variants play a role in risk for autism? Evidence and theoretical musings. Brain Res 1380: 78–84.
- Pinto D, Pagnamenta AT, Klei L, Anney R, Merico D, et al. (2010) Functional impact of global rare copy number variation in autism spectrum disorders. Nature 466: 368–72.
- Levy D, Ronemus M, Yamrom B, Lee Y, Leotta A, et al. (2011) Rare de novo and transmitted copy-number variation in autistic spectrum disorders. Neuron 70: 886–897.
- Sanders S, Hus V, Luo R, Murtha M, Moreno-De-Luca D, et al. (2011) Multiple recurrent *de novo* cnvs, including duplications of the 7q11. 23 williams syndrome region, are strongly associated with autism. Neuron 70: 863–885.
- O'Roak B, Deriziotis P, Lee C, Vives L, Schwartz J, et al. (2012) Exome sequencing in sporadic autism spectrum disorders identifies severe *de novo* mutations. Nature genetics 44: 471–471.
- Chahrour M, TimothyW, Lim E, Ataman B, Coulter M, et al. (2012)Wholeexome sequencing and homozygosity analysis implicate depolarization-regulated neuronal genes in autism. PLoS Genetics 8: e1002635.
- Kiezun A, Garimella K, Do R, Stitziel NO, Neale BM, et al. (2012) Exome sequencing and the genetic basis of complex traits. Nat Genet 44: 623–630.
- Pritchard JK (2001) Are rare variants responsible for susceptibility to complex diseases? Am J Hum Genet 69: 124–137.
   O'D', PLUE, PLUE,
- O'Roak BJ, Vives L, Fu W, Egertson JD, Stanaway IB, et al. (2012) Multiplex targeted sequencing identifies recurrently mutated genes in autism spectrum disorders. Science 338: 1619–1622.
- Lin MY, Lin YM, Kao TC, Chuang HH, Chen RH (2011) PDZ-RhoGEF ubiquitination by Cullin3-KLHL20 controls neurotrophin-induced neurite outgrowth. J Cell Biol 193: 985–994.
- Sobieszczuk DF, Poliakov A, Xu Q, Wilkinson DG (2010) A feedback loop mediated by degradation of an inhibitor is required to initiate neuronal differentiation. Genes Dev 24: 206–218.
- Schaefer H, Rongo C (2006) KEL-8 is a substrate receptor for CUL3-dependent ubiquitin ligase that regulates synaptic glutamate receptor turnover. Mol Biol Cell 17: 1250–1260.
- Kong A, Frigge ML, Masson G, Besenbacher S, Sulem P, et al. (2012) Rate of *de novo* mutations and the importance of father's age to disease risk. Nature 488: 471–5.

- Tong Y, Xu Y, Scearce-Levie K, Ptacek LJ, Fu YH (2010) COL25A1 triggers and promotes Alzheimer's disease-like pathology in vivo. Neurogenetics 11: 41– 52.
- Li D, Zhao H, Kranzler HR, Oslin D, Anton RF, et al. (2012) Association of COL25A1 with comorbid antisocial personality disorder and substance dependence. Biol Psychiatry 71: 733–740.
- Bedogni F, Hodge RD, Elsen GE, Nelson BR, Daza RA, et al. (2010) Tbr1 regulates regional and laminar identity of postmitotic neurons in developing neocortex. Proc Natl Acad Sci USA 107: 13129–13134.
- Angus SP, Nevins JR (2012) A role for Mediator complex subunit MED13L in Rb/E2F-induced growth arrest. Oncogene 31: 4709–4717.
- Ghanem N, Andrusiak MG, Svoboda D, Al Lafi SM, Julian LM, et al. (2012) The Rb/E2F pathway modulates neurogenesis through direct regulation of the Dlx1/Dlx2 bigene cluster. J Neurosci 32: 8219–8230.
- Andrusiak MG, McClellan KA, Dugal-Tessier D, Julian LM, Rodrigues SP, et al. (2011) Rb/E2F regulates expression of neogenin during neuronal migration. Mol Cell Biol 31: 238–247.
- Tomppo L, Ekelund J, Lichtermann D, Veijola J, Jarvelin MR, et al. (2012) DISC1 conditioned GWAS for psychosis proneness in a large Finnish birth cohort. PLoS ONE 7: e30643.
- Zheng S, Eacker SM, Hong SJ, Gronostajski RM, Dawson TM, et al. (2010) NMDA-induced neuronal survival is mediated through nuclear factor I-A in mice. J Clin Invest 120: 2446–2456.
- Shu T, Butz KG, Plachez C, Gronostajski RM, Richards LJ (2003) Abnormal development of forebrain midline glia and commissural projections in Nfia knock-out mice. J Neurosci 23: 203–212.
- Gordon D, Heath SC, Liu X, Ott J (2001) A transmission/disequilibrium test that allows for genotyping errors in the analysis of single-nucleotide polymorphism data. Am J Hum Genet 69: 371–380.
- Kim S, Millard SP, Yu CE, Leong L, Radant A, et al. (2012) Inheritance model introduces differential bias in CNV calls between parents and offspring. Genet Epidemiol 36: 488–498.
- Leek JT, Scharpf RB, Bravo HC, Simcha D, Langmead B, et al. (2010) Tackling the widespread and critical impact of batch effects in high-throughput data. Nat Rev Genet 11: 733–739.
- Tennessen JA, Bigham AW, O'Connor TD, FuW, Kenny EE, et al. (2012) Evolution and functional impact of rare coding variation from deep sequencing of human exomes. Science 337: 64–69.